Accelerator Status Report

Run Schedule: Jan. 3, 2011 through May 13, 2011
Jan. 26, 2011
Injector    Energy 61.76 MeV
· (1/25) Injector not stable with high current (>100uA) starting owl shift. Something is not stable but we have not identified the source yet. We were able to get beam current up to 180uA by adjusting Hall C laser phase and pre-buncher settings.
· (1/24) Laser spot moved to boost up current for Hall C. 
· (1/18) The alignment of the injector chicane was corrected. It sagged about 10mm.

· (1/18) Pre-buncher amplifier was replaced to stabilize beam.

· High current (>100uA) to Hall C is not stable: 0L03 or 0L04 trips. The experts have been investigating. 
Accelerator   Linac Energy 549.00 MeV
· The accelerator is stable.

· (1/24) The CHL tripped. 

· (1/23) The ESR tripped.
· (1/22) Opened up the injector and the NL for the USPAS tour. Box supplies of Arc8 and Arc2, and vacuum in the NL delayed the restoration of the beam for a few hours.
· The optics in the main machine has been stable although the FOPT data showed that the betatron match at Arc 5 changed a bit.
· (1/18) The pre-buncher amplifier failed during the injector beam tuning. 

· PSS key C2 does not work and it has been tagged out. Do not use this key.

· 2L19-7 CM replaced.

Hall A---
· Will not be ready for beam until Feb. 11 due to installation of a new experiment.
Hall B ---Pass 5, TPE
· Hall put more shielding to protect its nA BPMs electronics, which appears to be working.

· nA BPMs had problems on a few occasions, which caused some down time. The ioc was rebooted a few times. The amplifier was replaced a couple of times. (~9 hrs.)
Hall C--- Pass 1, QWeak 
· It seems that something in the injector is not stable, which causes the high current beam to unstable.

· High current to the hall is not stable sometimes.

